UCLA UCLA Previously Published Works

Title

Variation in benefit among patients with serious mental illness who receive integrated psychiatric and primary care.

Permalink

https://escholarship.org/uc/item/0z88v8jz

Journal PLoS ONE, 19(5)

Authors

Skela, Jessica Chang, Evelyn Oberman, Rebecca <u>et al.</u>

Publication Date

2024

DOI

10.1371/journal.pone.0304312

Peer reviewed



G OPEN ACCESS

Citation: Young AS, Skela J, Chang ET, Oberman R, Siddarth P (2024) Variation in benefit among patients with serious mental illness who receive integrated psychiatric and primary care. PLoS ONE 19(5): e0304312. https://doi.org/10.1371/journal.pone.0304312

Editor: Mu-Hong Chen, Taipei Veterans General Hospital, TAIWAN

Received: January 21, 2024

Accepted: May 6, 2024

Published: May 23, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0304312

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: "The data used in this study are the property of the United States government Department of Veterans Affairs. Electronic health records contain substantial RESEARCH ARTICLE

Variation in benefit among patients with serious mental illness who receive integrated psychiatric and primary care

Alexander S. Young^{1,2,3}*, Jessica Skela², Evelyn T. Chang^{3,4,5}, Rebecca Oberman³, Prabha Siddarth²

 Desert Pacific Mental Illness Research Education and Clinical Center, Greater Los Angeles Veterans Healthcare System, Los Angeles, California, United States of America, 2 Department of Psychiatry, School of Medicine, University of California Los Angeles, Los Angeles, California, United States of America, 3 HSR&D Center for the Study of Healthcare Innovation, Implementation and Policy, Greater Los Angeles Veterans Healthcare System, Los Angeles, California, United States of America, 4 Department of Medicine, Veterans Affairs Greater Los Angeles Healthcare System, Los Angeles, California, United States of America,
Department of Medicine, David Geffen School of Medicine, University of California Los Angeles, Los Angeles, California, United States of America

* ayoung@ucla.edu

Abstract

Purpose

The population with serious mental illness has high risk for hospitalization or death due to unhealthy behaviors and inadequate medical care, though the level of risk varies substantially. Programs that integrate medical and psychiatric services improve outcomes but are challenging to implement and access is limited. It would be useful to know whether benefits are confined to patients with specific levels of risk.

Methods

In a population with serious mental illness and increased risk for hospitalization or death, a specialized medical home integrated services and improved treatment and outcomes. Treatment quality, chronic illness care, care experience, symptoms, and quality of life were assessed for a median of 385 days. Analyses examine whether improvements varied by baseline level of patient risk.

Results

Patients with greater risk were more likely to be older, more cognitively impaired, and have worse mental health. Integrated services increased appropriate screening for body mass index, lipids, and glucose, but increases did not differ significantly by level of risk. Integrated services also improved chronic illness care, care experience, mental health-related quality of life, and psychotic symptoms. There were also no significant differences by risk level.

sensitive information, private information and information that is protected by United States statute and regulation, including HIPAA. Therefore, it is not possible for us to share the data used in these analyses. Data can be requested from the authors or the research service of the Great Los Angeles Veterans Healthcare System, which can be reached at 1-310-268-4437 or

VHAWLAResearchSvc@va.gov".

Funding: This research was supported by the U.S. Department of Veterans Affairs, Veterans Health Administration, Health Services Research and Development Service Quality Enhancement Research Initiative (SDP 12-177; AY), and Desert Pacific Mental Illness Research, Education and Clinical Center (MIRECC; AY). https://www.queri. research.va.gov/ Funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions

There were benefits from integration of primary care and psychiatric care at all levels of increased risk, including those with extremely high risk above the 95th percentile. When developing integrated care programs, patients should be considered at all levels of risk, not only those who are the healthiest.

Introduction

Healthcare organizations are increasingly responsible for providing comprehensive care of populations of patients, and for improving treatment quality while controlling costs. Serious mental illnesses such as bipolar disorder or schizophrenia are present in a substantial proportion of patients with preventable emergency visits, hospitalizations or death [1,2]. A major cause of increased risk is inadequate primary care for common medical conditions [3–6]. There are interventions that integrate psychiatric and primary care and improve outcomes, but these require additional resources and reorganization of care, and access to these services is limited [7–10]. Patients with high risk for poor outcomes, including those with serious mental illness (SMI) have the most potential to benefit from these interventions. However, it is not clear whether care integration efforts benefit all patients with increased health risk, or whether these services can be focused more narrowly.

At a large Veterans Health Administration (VHA) healthcare system, psychiatric care was integrated into a primary care medical home that focused on patients with SMI and increased risk. The VHA refers to medical homes as Patient Aligned Care Teams (PACTs), and this medical home was referred to as SMI PACT [11]. In a prospective controlled trial, SMI PACT improved treatment experience, quality of care, and patient outcomes [8]. The program selected eligible patients for enrollment from individuals with increased risk. Risk was quantified using the Care Assessment Need (CAN) score, which the VHA calculates on a weekly basis for all patients [12]. The CAN score accurately estimates the risk of hospitalization or death using a model that includes many variables drawn from the domains of demographics, diagnoses, vital signs, medications, laboratory values and service utilization. Expressed as a percentile of risk, the CAN one-year risk for hospitalization or death increases from about 15% at the 75th percentile to 51% for all who are above the 95th percentile of risk [12].

Organizations have used risk prediction to focus enhanced service delivery on patients with high risk for poor outcomes [13–16]. Patients with greater risk would be expected to have the potential for greater improvement with enhanced programs or new interventions. However, they might also be more challenging to engage and treat, and very high risk may not be modifiable. There has been little understanding of how baseline risk contributes to improvement in treatment and outcomes with integration of care. It would be useful to know whether all patients benefit, or whether enhanced programs can be focused more narrowly. To inform future efforts to target service delivery, this manuscript examines whether improvements in treatment and outcomes vary by patients' level of risk.

Methods

Design

The SMI PACT project was registered with ClinicalTrials.gov (identifier NCT01668355) and the research methods and clinical trial results have been published previously [8,11,17]. Service

integration occurred at the Veterans Affairs Greater Los Angeles Healthcare System, one of the largest VHA healthcare systems in the United States. Patients were eligible if they had: 1) a CAN score over the 75th percentile of risk, and 2) SMI, which was defined as having a diagnosis of schizophrenia, schizoaffective disorder, bipolar disorder, recurrent major depression with psychosis, or chronic severe post-traumatic stress disorder requiring antipsychotic medication treatment. From 1,322 eligible patients, a sample of 164 were randomly selected and offered care from the SMI PACT medical home. This sample size was determined using power analysis regarding the number required to detect change in treatment utilization. Patients were recruited between May 3, 2016, and February 21, 2018. This study was approved by the IRB of the Veterans Affairs Greater Los Angeles Healthcare System. Participants provided written informed consent.

Data are drawn from in-person assessments that were performed by trained research staff who were blind to patients' CAN scores. Measures include gender, race, ethnicity, education attainment, psychiatric diagnosis, level of mental health recovery (MORS) [18], cognitive functioning in the domains of verbal learning and speed of processing (HVLT-R and digit symbol substitution test) [19,20], psychotic symptom severity (BASIS-R psychosis scale) [21], ambulatory care experience (ACES) [22], receipt of chronic illness care (PACIC) [23], patient activation (PAM-13) [24], and health related mental (MCS) and physical (PCS) components of quality of life (VR-12) [25]. VHA administrative information systems provided data on lab test results, height, weight, blood pressure, diagnoses, prescriptions, and services. These data were used to calculate Body Mass Index (BMI), and measures of treatment appropriateness and quality, including measures of metabolic screening and monitoring based on specifications from NCQA, HEDIS, and VA.

Analysis

Patients were grouped into risk strata based on CAN percentile score at baseline. Demographics and medical record data were compared among risk groups using chi square tests for categorical variables and analysis of variance for continuous variables. These were univariate tests (one variable at a time). A cumulative logistic regression model was used to estimate the effect of individual patient characteristics on health risk. Odds ratios were calculated to assess the contribution of each variable associated with health risk. Goodness of fit was determined by calculating the area under the receiver operating characteristic curve. To examine the effect of health risk on treatment quality and outcomes, linear mixed effects models with repeated measures were estimated. Separate models were run for each of the continuous outcomes, with CAN group, time (baseline and post-intervention), and group x time interaction as predictors. Generalized linear mixed effects models were used for the binary outcomes (metabolic screening measures), with the same predictors of CAN group, time, and group x time interaction. Statistical significance was defined as a 2-sided p-value < .05.

Results

Table 1 describes the sample, including characteristics and in-person assessment measures stratified by CAN risk percentile of 75–85 (n = 89), 85–95 (n = 51) and 95–99 (n = 23). Nearly 80% of patients had some college education, 35% had schizophrenia, 35% had bipolar disorder, and 28% had chronic, severe, disabling Post-Traumatic Stress Disorder. Cognitive functioning and health-related quality of life were substantially below general population means (mean HVLT-R = 5.1, SD = 1.8; mean Digit Symbol Test = 38.4, SD = 10.9; mean PCS = 36.4, SD = 11.1; mean MCS = 39.9, SD = 13.1). Greater health risk was significantly associated with advancing age (odds ratio per decade = 1.49, p = 0.02) and worse mental health-related quality

Characteristic	Overall (N = 164)	CAN 75-85 (N = 89)	CAN 85-95 (N = 51)	CAN 95-99 (N = 23)	
	N (%) or Mean±SD	N (%) or Mean±SD	N (%) or Mean±SD	N (%) or Mean±SD	p-value
Gender					0.93
Female	14 (8.5)	7 (7.9)	5 (9.8)	2 (8.7)	
Male	150 (91.5)	82 (92.1)	46 (90.2)	21 (91.3)	
Age, years	58.7 ± 10.4	57.3 ± 11.1	59.7 ± 8.7	63.2 ± 7.7	0.01
Race					0.63
White	59 (36.2)	35 (39.8)	16 (31.4)	8 (34.8)	
Black	74 (45.4)	38 (43.2)	27 (52.9)	9 (39.1)	
Other	30 (18.4)	15 (17.0)	8 (15.7)	6 (26.1)	
Ethnicity					0.89
Hispanic or Latino	22 (13.7)	11 (12.6)	7 (14.3)	3 (13.0)	
Not Hispanic or Latino	138 (86.3)	76 (87.4)	42 (85.7)	20 (87.0)	
Education					0.76
Less than high school	7 (4.3)	3 (3.4)	3 (5.9)	1 (4.3)	
High school or equivalent	26 (15.9)	18 (20.2)	5 (9.8)	3 (13.0)	
Some college	115 (70.1)	60 (67.4)	38 (74.5)	16 (69.7)	
Some graduate school	16 (9.7)	8 (9.0)	5 (9.8)	3 (13.0)	
Diagnosis					0.40
Schizophrenia	57 (34.8)	28 (31.5)	20 (39.2)	9 (39.1)	
Bipolar disorder	57 (34.8)	34 (38.2)	18 (35.3)	4 (17.4)	
Major depression with psychosis	4 (2.4)	1 (1.1)	2 (3.9)	1 (4.3)	
Chronic disabling PTSD	46 (28.0)	26 (29.2)	11 (21.6)	9 (39.1)	
Recovery, MORS	6.4 ± 0.6	6.4 ± 0.6	6.3 ± 0.6	6.4 ± 0.7	0.52
Cognition, HVLT-R	5.1 ± 1.8	5.2 ± 1.7	5.3 ± 1.7	4.3 ± 1.9	0.12
Cognition, Digit Symbol	38.4 ± 10.9	39.5 ± 10.8	38.6 ± 11.1	33.1 ± 9.5	0.03
Chronic illness care	2.7 ± 1.0	2.6 ± 1.0	2.82 ± 1.1	2.54 ± 1.0	0.97
Ambulatory care experience	68.1 ± 20.5	66.8 ± 18.8	76.7 ± 20.3	57. ± 25.6	0.85
Patient activation	2.1 ± 0.4	2.2 ± 0.4	2.1 ± 0.5	1.9 ± 0.4	0.07
Physical quality of life	36.4 ± 11.1	37.7 ± 12.0	33.5 ± 9.5	36.8 ± 9.5	0.26
Mental health quality of life	39.9 ± 13.1	41.8 ± 13.0	39.6 ± 12.8	34.3 ± 12.7	0.02
Psychotic symptoms	0.9 ± 0.8	0.8 ± 0.7	1.0 ± 0.9	0.9 ± 0.8	0.20

Table 1. The characteristics of patients stratified by level of health risk.

https://doi.org/10.1371/journal.pone.0304312.t001

of life (odds ratio = 0.93, p < .001). Risk was not significantly associated with gender, race, ethnicity, educational level, specific psychiatric diagnosis, cognition, psychiatric symptoms, physical health related quality of life, chronic illness care, experience of care, or patient activation. Diagnostic accuracy of this model was 72.3% (95% CI 65.8–78.8).

Participants who were assessed at follow-up (n = 133) participated in the intervention for a median of 385 days (mean = 427, SD = 106). Mean days of participation were 414 (SD = 97.9) for the 75–85 CAN group, 446 (SD = 116.9) for the 85–95 CAN group, and 441 (SD = 110.4) for the 95–99 CAN group. These three groups were not significantly different in their number of days of participation (Kruskal Wallis chi square (2) = 5.4, p = .07).

As previously reported, the SMI PACT intervention improved chronic illness care, care experience, mental health-related quality of life and psychotic symptoms, as well as screening for body mass index, lipids and glucose, compared to usual care [8]. Table 2 presents improvements in the quality of care with SMI PACT, stratified by risk group. There were improvements in the appropriate screenings for body mass index, lipids, and glucose, however, these did not vary by stratum of risk (Wald $\chi^2 = 0.0-1.40$, p = 0.23–1).

Measure	Before	After	β (CAN group	Chi-Square	p-value
	Medical Home	Medical Home	x time), p-value	(df)	
	n (%)	n (%)			
BMI Screening					
CAN Group 75–85	69 (77.5)	75 (84.3)		0.18 (1)	0.7
CAN Group 85–95	42 (82.4)	46 (90.2)	0.239, 0.7		
CAN Group 95–99	21 (91.3)	22 (95.7)	0.299, 0.8		
Blood Pressure Screening	·				
CAN Group 75–85	72 (80.9)	74 (83.1)		0 (1)	1
CAN Group 85–95	48 (94.1)	48 (94.1)	-0.153, 0.9		
CAN Group 95–99	23 (100.0)	23 (100.0)	-0.153, 0.9		
Lipid Screening					
CAN Group 75–85	33 (37.1)	42 (47.2)		1.4 (1)	0.2
CAN Group 85–95	12 (23.5)	31 (60.8)	1.201, 0.03		
CAN Group 95–99	7 (30.4)	15 (65.2)	1.034, 0.14		
Glucose Screening					
CAN Group 75–85	46 (51.7)	53 (59.6)		0.52 (1)	0.5
CAN Group 85–95	35 (68.6)	44 (86.3)	0.736, 0.2		
CAN Group 95–99	19 (82.6)	21 (91.3)	0.474. 0.6		

Table 2.	The effect of inte	egrated care on	treatment quali	ty stratified b	y baseline risk.

https://doi.org/10.1371/journal.pone.0304312.t002

Table 3 presents improvements in patient outcomes, stratified by risk group. While there were improvements in chronic illness care, care experience, mental health related quality of life, and psychotic symptoms, improvements also did not vary significantly by risk (F = 0.21-2.86, p = 0.06-0.8). Even the stratum with extremely high risk (> 95^{th} percentile) improved in chronic illness care and physical health related quality of life.

Discussion

This study suggests that improvements in care and outcomes do not vary significantly by level of health risk. There is improvement in healthcare and outcomes across multiple domains even for those with extremely high risk. However, the magnitude of improvement can vary. With this study's intervention, there was minimal improvement in patient activation or physical health-related quality of life. While there was substantial improvement in the other outcomes at each level of risk, psychotic symptoms did not improve for the group with the highest level of risk. While not all people with SMI have increased health risk, there is a large population at high risk of hospitalization or death. Using risk prediction, this project identified the population of patients with SMI and high risk for hospitalization or death. Patients with greater health risk were more likely to be older, more cognitively impaired, and have worse mental health. There were benefits from integration of primary care and psychiatric care at all levels of increased risk, including those with risk above the 95th percentile. The patients with SMI were very receptive to medical preventive services [26], and substantial improvements were seen in some measures of screening, such as for lipids.

A potential limitation of this study is that it was conducted in the VHA. The VHA offers comprehensive healthcare services and provides people with SMI better access to medical care than some other organizations. The health risk of people with SMI could be greater in systems with more problems in access to care. Also, while the sample size in this study was based on an a priori power calculation, it is possible that we did not have enough power to detect small differences in outcomes among the three risk groups.

Measure	Before	Fore After Change		Statis	Statistic	
	Medical Home	Medical Home				
	mean (SD)	mean (SD)	mean (SD), p-value	F (ndf,ddf)	p-value	
Chronic illness care				1.9 (2, 115)	0.15	
CAN 75-85	2.5 (0.9)	3.6 (1.0)	0.98 (1.2), < 0.001			
CAN 85-95	2.8 (1.0)	3.5 (0.8)	0.69 (0.9), < 0.001			
CAN 95-99	2.4 (1.0)	2.9 (0.9)	0.66 (0.9), 0.05			
Care experience				0.32 (2, 138)	0.7	
CAN 75-85	70.3 (23.8)	85.3 (17.4)	13.25 (22.9), <0.001			
CAN 85-95	76.7 (22.2)	87.0 (13.1)	11.07 (23.6), 0.01			
CAN 95-99	69.3 (26.6)	81.4 (21.0)	15.15 (28.3), 0.02			
Patient activation						
CAN 75-85	2.2 (0.4)	2.2 (0.5)	0.05 (0.4), 0.3	0.56 (2, 139)	0.6	
CAN 85-95	2.1 (0.5)	2.3 (0.5)	0.11 (0.4), 0.06			
CAN 95-99	1.90 (0.4)	1.9 (0.4)	0.02 (0.3), 0.8			
Physical health-related quality of life				2.86 (2, 138)	0.06	
CAN 75-85	37.7 (12.0)	39.2 (11.5)	0.77 (9.2), 0.3			
CAN 85-95	33.5 (9.5)	34.9 (9.1)	0.50 (9.1), 0.6			
CAN 95-99	36.8 (9.5)	32.0 (9.2)	-3.97 (9.3), 0.04			
Mental health-related quality of life				0.21 (2, 140)	0.8	
CAN 75-85	41.8 (13.0)	44.2 (12.6)	2.37 (12.1), 0.08			
CAN 85-95	39.5 (12.8)	43.1 (12.5)	3.18 (10.5), 0.07			
CAN 95-99	34.3 (12.7)	38.0 (11.9)	4.38 (13.6), 0.12			
Psychotic symptoms						
CAN 75-85	0.8 (0.7)	0.6 (0.7)	-0.15 (0.7), 0.04	0.36 (2,140)	0.7	
CAN 85-95	1.0 (0.9)	0.8 (0.8)	-0.24 (0.8), 0.05			
CAN 95-99	0.9 (0.8)	0.9 (0.8)	-0.09 (0.7), 0.7			

Table 3. The effect of integrated care on patient outcomes stratified by baseline risk.

https://doi.org/10.1371/journal.pone.0304312.t003

Studies have found that health can be improved in populations with SMI with interventions that integrate medical and psychiatric care [27–29]. The large disparities in health and health-care seen in this population can be improved. The routine use of health risk assessment can support the implementation and conduct of interventions to improve medical care of people with increased health risk and can strengthen such models. In the context of limited resources, it is reasonable to focus integration of care on patients with increased risk for poor outcomes since there is the most potential benefit in these patients. Within this population, there is similar benefit for patients at each level of increased risk, including those with extremely high risk for poor outcomes. Integrated care programs should consider recruiting all patients with health risk, not only those who are the heathiest.

Acknowledgments

The contents of this publication and the views expressed herein do not necessarily represent the views of the Department of Veterans Affairs or affiliated institutions.

Author Contributions

Conceptualization: Alexander S. Young, Evelyn T. Chang.

Data curation: Alexander S. Young, Jessica Skela, Prabha Siddarth.

Formal analysis: Jessica Skela, Prabha Siddarth.

Funding acquisition: Alexander S. Young.

Investigation: Alexander S. Young, Evelyn T. Chang, Rebecca Oberman.

Methodology: Alexander S. Young.

Project administration: Alexander S. Young, Rebecca Oberman.

Resources: Alexander S. Young.

Supervision: Alexander S. Young, Rebecca Oberman.

Writing - original draft: Alexander S. Young.

Writing – review & editing: Alexander S. Young, Jessica Skela, Evelyn T. Chang, Prabha Siddarth.

References

- Schmidt EM, Behar S, Barrera A, Cordova M, Beckum L. Potentially Preventable Medical Hospitalizations and Emergency Department Visits by the Behavioral Health Population. J Behav Health Serv Res. 2018; 45(3):370–88. Epub 2017/09/15. https://doi.org/10.1007/s11414-017-9570-y PMID: 28905296.
- Young AS, Chinman MJ, Cradock-O'Leary JA, Sullivan G, Murata D, Mintz J, et al. Characteristics of individuals with severe mental illness who use emergency services. Community Ment Health J. 2005; 41(2):159–68. Epub 2005/06/25. https://doi.org/10.1007/s10597-005-2650-0 PMID: 15974496.
- Ward MC, Druss BG. Treatment Considerations in Severe Mental Illness: Caring for the Whole Patient. JAMA Psychiatry. 2019; 76(7):759–60. Epub 2019/05/16. https://doi.org/10.1001/jamapsychiatry.2019. 0903 PMID: 31090878.
- Cradock-O'Leary J, Young AS, Yano EM, Wang M, Lee ML. Use of general medical services by VA patients with psychiatric disorders. Psychiatr Serv. 2002; 53(7):874–8. Epub 2002/07/04. https://doi. org/10.1176/appi.ps.53.7.874 PMID: 12096172.
- Iyer SP, Young AS. Health screening, counseling, and hypertension control for people with serious mental illness at primary care visits. Gen Hosp Psychiatry. 2015; 37(1):60–6. Epub 2014/12/07. https://doi.org/10.1016/j.genhosppsych.2014.10.003 PMID: 25480463.
- Janney CA, Greenberg JM, Moin T, Kim HM, Holleman RG, Hughes M, et al. Does mental health influence weight loss in adults with prediabetes? Findings from the VA Diabetes Prevention Program. Gen Hosp Psychiatry. 2018; 53:32–7. Epub 2018/05/08. https://doi.org/10.1016/j.genhosppsych.2018.04. 001 PMID: 29729586.
- Chang ET, Young AS. The role of medical homes in primary care. In: Corrigan PW, Ballentine S, editors. Health and Wellness in People Living with Serious Mental Illness. Washington, DC: American Psychiatric Association Press; 2021. p. 149–62.
- Young AS, Chang ET, Cohen AN, Oberman R, Chang DT, Hamilton AB, et al. The Effectiveness of a Specialized Primary Care Medical Home for Patients with Serious Mental Illness. J Gen Intern Med. 2022; 37(13):3258–65. Epub 2022/04/06. https://doi.org/10.1007/s11606-021-07270-x PMID: 35380346; PubMed Central PMCID: PMC9550991.
- Druss BG, Singh M, von Esenwein SA, Glick GE, Tapscott S, Tucker SJ, et al. Peer-Led Self-Management of General Medical Conditions for Patients With Serious Mental Illnesses: A Randomized Trial. Psychiatr Serv. 2018; 69(5):529–35. Epub 2018/02/02. https://doi.org/10.1176/appi.ps.201700352 PMID: 29385952; PubMed Central PMCID: PMC5930018.
- Lewis VA, Colla CH, Tierney K, Van Citters AD, Fisher ES, Meara E. Few ACOs pursue innovative models that integrate care for mental illness and substance abuse with primary care. Health Aff (Millwood). 2014; 33(10):1808–16. Epub 2014/10/08. https://doi.org/10.1377/hlthaff.2014.0353 PMID: 25288427.
- Young AS, Cohen AN, Chang ET, Flynn AWP, Hamilton AB, Oberman R, et al. A clustered controlled trial of the implementation and effectiveness of a medical home to improve health care of people with serious mental illness: study protocol. BMC Health Serv Res. 2018; 18(1):428. Epub 2018/06/09. https://doi.org/10.1186/s12913-018-3237-0 PMID: 29880047; PubMed Central PMCID: PMC5992687.
- Wang L, Porter B, Maynard C, Evans G, Bryson C, Sun H, et al. Predicting risk of hospitalization or death among patients receiving primary care in the Veterans Health Administration. Med Care. 2013; 51(4):368–73. https://doi.org/10.1097/MLR.0b013e31827da95a PMID: 23269113.

- Yoon J, Chang E, Rubenstein LV, Park A, Zulman DM, Stockdale S, et al. Impact of Primary Care Intensive Management on High-Risk Veterans' Costs and Utilization: A Randomized Quality Improvement Trial. Ann Intern Med. 2018; 168(12):846–54. Epub 2018/06/06. https://doi.org/10.7326/M17-3039 PMID: 29868706.
- Nelson KM, Chang ET, Zulman DM, Rubenstein LV, Kirkland FD, Fihn SD. Using Predictive Analytics to Guide Patient Care and Research in a National Health System. J Gen Intern Med. 2019; 34(8):1379– 80. Epub 2019/04/24. https://doi.org/10.1007/s11606-019-04961-4 PMID: 31011959; PubMed Central PMCID: PMC6667597.
- Wong ES, Yoon J, Piegari RI, Rosland AM, Fihn SD, Chang ET. Identifying Latent Subgroups of High-Risk Patients Using Risk Score Trajectories. J Gen Intern Med. 2018; 33(12):2120–6. Epub 2018/09/ 19. https://doi.org/10.1007/s11606-018-4653-x PMID: 30225769; PubMed Central PMCID: PMC6258600.
- Chang ET, Raja PV, Stockdale SE, Katz ML, Zulman DM, Eng JA, et al. What are the key elements for implementing intensive primary care? A multisite Veterans Health Administration case study. Healthc (Amst). 2018; 6(4):231–7. Epub 2017/11/06. <u>https://doi.org/10.1016/j.hjdsi.2017.10.001</u> PMID: 29102480.
- Chang ET, Vinzon M, Cohen AN, Young AS. Effective Models Urgently Needed to Improve Physical Care for People With Serious Mental Illnesses. Health Serv Insights. 2019; 12:1178632919837628. Epub 2019/05/30. https://doi.org/10.1177/1178632919837628 PMID: <u>31138983</u>; PubMed Central PMCID: PMC6518543.
- Fisher DG, Pilon D, Hershberger SL, Reynolds GL, LaMaster SC, Davis M. Psychometric properties of an assessment for mental health recovery programs. Community Ment Health J. 2009; 45(4):246–50. Epub 2009/07/08. https://doi.org/10.1007/s10597-009-9213-8 PMID: 19582574; PubMed Central PMCID: PMC2718196.
- Cherner M, Suarez P, Lazzaretto D, Fortuny LA, Mindt MR, Dawes S, et al. Demographically corrected norms for the Brief Visuospatial Memory Test-revised and Hopkins Verbal Learning Test-revised in monolingual Spanish speakers from the U.S.-Mexico border region. Arch Clin Neuropsychol. 2007; 22 (3):343–53. Epub 2007/02/13. https://doi.org/10.1016/j.acn.2007.01.009 PMID: 17293078; PubMed Central PMCID: PMC3737073.
- Smith A. Symbol Digit Modalities Test (SDMT). Manual (Revised). Los Angeles CA: Western Psychological Services; 1982.
- Eisen SV, Normand SL, Belanger AJ, Spiro A, III, Esch D. The Revised Behavior and Symptom Identification Scale (BASIS-R): reliability and validity. Med Care. 2004; 42(12):1230–41. <u>https://doi.org/10.1097/00005650-200412000-00010</u> PMID: 15550803.
- 22. Safran DG, Karp M, Coltin K, Chang H, Li A, Ogren J, et al. Measuring patients' experiences with individual primary care physicians. J Gen Intern Med. 2006; 21(1):13–21. <u>https://doi.org/10.1111/j.1525-1497.2005.00311.x PMID: 16423118</u>.
- Glasgow RE, Wagner EH, Schaefer J, Mahoney LD, Reid RJ, Greene SM. Development and validation of the Patient Assessment of Chronic Illness Care (PACIC). Med Care. 2005; 43(5):436–44. Epub 2005/04/20. https://doi.org/10.1097/01.mlr.0000160375.47920.8c PMID: 15838407.
- 24. Hibbard JH, Mahoney ER, Stockard J, Tusler M. Development and testing of a short form of the patient activation measure. Health Serv Res. 2005; 40(6 Pt 1):1918–30. Epub 2005/12/13. https://doi.org/10. 1111/j.1475-6773.2005.00438.x PMID: 16336556; PubMed Central PMCID: PMC1361231.
- Selim AJ, Rogers W, Fleishman JA, Qian SX, Fincke BG, Rothendler JA, et al. Updated U.S. population standard for the Veterans RAND 12-item Health Survey (VR-12). Qual Life Res. 2009; 18(1):43–52. Epub 2008/12/04. https://doi.org/10.1007/s11136-008-9418-2 PMID: 19051059.
- Bergman AA, Chang ET, Cohen AN, Hovsepian S, Oberman RS, Vinzon M, et al. Patient experiences with a primary care medical home tailored for people with serious mental illness. Fam Syst Health. 2023. Epub 20231113. https://doi.org/10.1037/fsh0000853 PMID: 37956066.
- Druss BG, Rohrbaugh RM, Levinson CM, Rosenheck RA. Integrated medical care for patients with serious psychiatric illness: a randomized trial. Arch Gen Psychiatry. 2001; 58(9):861–8. Epub 2001/09/26. https://doi.org/10.1001/archpsyc.58.9.861 PMID: 11545670.
- Bradford DW, Cunningham NT, Slubicki MN, McDuffie JR, Kilbourne AM, Nagi A, et al. An evidence synthesis of care models to improve general medical outcomes for individuals with serious mental illness: a systematic review. J Clin Psychiatry. 2013; 74(8):e754–64. Epub 2013/09/12. https://doi.org/10. 4088/JCP.12r07666 PMID: 24021516.
- Druss BG, von Esenwein SA, Glick GE, Deubler E, Lally C, Ward MC, et al. Randomized trial of an integrated behavioral health home: The Health Outcomes Management and Evaluation (HOME) study. Am J Psychiatry. 2017; 174(3):246–55. https://doi.org/10.1176/appi.ajp.2016.16050507 PMID: 27631964.